Author’s response to reviews

Title: Depression, Smoking, Physical Inactivity and Season independently associated with Midnight Salivary Cortisol in Type 1 Diabetes

Authors:

Eva O Melin (eva.melin@ltkronoberg.se)
Maria Thunander (maria.thunander@ltkronoberg.se)
Mona Landin-Olsson (mona.landin-olsson@med.lu.se)
Magnus Hillman (magnus.hillman@med.lu.se)
Hans O Thulesius (hansthulesius@gmail.com)

Version: 2
Date: 21 May 2014

Author’s response to reviews:

To BMC Endocrine Disorders, 21/05/2014.

Dear professor Timothy Shipley, editor-in-chief of BMC Endocrine Disorders,

We hereby submit two corrections in the manuscript “Depression, Smoking, Physical Inactivity and Season independently associated with Midnight Salivary Cortisol in Type 1 Diabetes” by authors Eva O. Melin, Maria Thunander, Mona Landin-Olsson, Magnus Hillman, Hans O. Thulesius.

First, at line 42 the correct is HbA1c >70 mmol/mol, not 70 mmol/L.

Second, at line 312 we changed into circadian rhythm from diurnal rhythm.

We still hope that our article will be considered for publication as an original research article in BMC Endocrine Disorders.

Sincerely,

Eva Melin, MD, PhD student (defending PhD later this year).
Endocrinology and Diabetes, Department of Clinical Sciences, Lund University, Lund.
Department of Research and Development, Kronoberg County Council, Växjö.
Primary Care, Kronoberg County Council, Växjö, Sweden.